Author's response to reviews

Title: Indigenous well-being in four countries: An application of the UNDP'S Human Development Index to Indigenous Peoples in Australia, Canada, New Zealand, and the United States

Authors:

Martin Cooke (cooke@uwaterloo.ca)
Francis Mitrou (francism@ichr.uwa.edu.au)
David Lawrence (D.Lawrence@curtin.edu.au)
Eric Guimond (guimonde@ainc-inac.gc.ca)
Dan Beavon (beavond@ainc-inac.gc.ca)

Version: 5 Date: 27 November 2007

Author's response to reviews:

Thank you for accepting our paper for publication. The following changes have been made to the manuscript as per your formatting instructions:

Title - Please use sentence case in the title, both in the manuscript file and the submission system (i.e. remove all unnecessary capitals).

The title is in correct sentence case. Human Development Index is capitalised as a proper name, country names of course are capitalised, and Indigenous Peoples is capitalised as a traditional mark of respect.

Abstract - Please ensure that the Abstract in the manuscript matches exactly what is in the submission system.

Done.

Table and figure placement text - Please remove this.

These have been removed.

Section headings - Please use sentence case for all headings and sub-headings in the manuscript (i.e. remove all unnecessary capitals). For example Results and discussion not Results and Discussion.

Done.

Table shading - We are unable to display shading and colours within the tables. Can we therefore ask you to use another method of highlighting the shaded text e.g. by making it bold.

Done.

We hope the manuscript will now be satisfactory for publication. If you have any
additional requests, please do not hesitate to contact us.